University of New Brunswick, Saint John Math 1013 Sections 2B, 3B Intoduction to Calculus II Winter 2006

Location and Time: 2B: GH215: MTWF 11:30-12:20 3B: HH239: MTWF 13:30-14:20 Instructor: Trevor Jones Office: Annex A, A13 Course Website: http://tjones1.unbsj.ca E-mail: tjones1@unbsj.ca Must put M1013 2B or M1013 3B in the subject line.

Note: Free tutoring is available through the Math Help Center in Oland Hall, G11. Prerequisite: A grade of C or better in Math 1003 Text: Calculus: Early Transcendentals, James Stewart, 5^{th} edition Syllabus: We will attempt to cover the following chapters of the text, though not necessarily in this order: Chapter 5, 6, 7, 8, 9, 17

Assignments Assignments must be handed in by the beginning of the class on the day due. and Quizzes: 10% Late assignments will not be marked. The assignments must be neatly written, in blue or black ink, on 8.5”x11” paper, single sided, and stapled in the upper left-hand corner. The assignment questions must be clearly labeled. You must include a title page which contains your name and student number, the course and section number, and the assignment number. Marks will be deducted for assignments failing to meet these requirements.

There will be pop quizzes during class time. These will be short, open book, open notes, calculator allowed quizzes.

Tests: 40% There will be four tests, each worth 10%. The tests take place on January 24^{th}, February 10^{th}, March 1^{st}, and March 22^{nd}.

There will be no make-up tests. If you have a valid written excuse, marks for the test will be redistributed. No electronic devices will be permitted during the tests. You must bring your student ID to all tests.

Exam: 50% No electronic devices will be permitted during the exam. You must bring your student ID to the exam.